Author’s response to reviews

Title: Medical students: what educational resources are they using?

Authors:

Lucinda Wynter (lucy.wynter@sydney.edu.au)

Annette Burgess (annette.burgess@sydney.edu.au)

Jack Heron (jheron@gmail.com)

Jane Bleasel (jane.bleasel@sydney.edu.au)

Version: 1 Date: 21 Feb 2018

Author’s response to reviews:

Dear Editor,

thank you for pointing out that we had not uploaded Figure 1. This has now been corrected.

Regards,

Annette